Author's response to reviews

Title: Cognitive function and quality of life in multiple sclerosis patients: a cross-sectional study

Authors:

Karine Baumstarck-Barrau (karine.baumstarck@univmed.fr)
Marie-Claude Simeoni (marie-claude.simeoni@univmed.fr)
Francoise Reuter (francoise.reuter@inserm.fr)
Irina Klemina (irina.klemina@ap-hm.fr)
Valerie Aghababian (valerie.aghanabian@univmed.fr)
Jean Pelletier (jean.pelletier@ap-hm.fr)
Pascal Auquier (pascal.auquier@univmed.fr)

Version: 4 Date: 22 December 2010

Author's response to reviews:

December 22, 2010
Melissa Norton, Editor-in-Chief
BMC Neurology editorial

Dear Jelena Drulovic and Prof. Robert Zivadinov,


We thank you for accepting a new version of our manuscript. We are grateful to the reviewers for their examination of our revised manuscript and their constructive comments/suggestions. We integrated their suggestions in this version. We highlighted the changes made in the manuscript (yellow color).

We hope that this new version will be acceptable.

Yours sincerely,

Karine Baumstarck-Barrau
Pascal Auquier
EA3279, Self-perceived Health Assessment Research Unit, School of Medicine, Université de la Méditerranée, 27 bd Jean Moulin, Marseille cedex 05, F-13385 France
Phone: +33 491 384 744, Fax: +33 491 384 482
E-mail: karine.baumstarck@univmed.fr, pascal.auquier@univmed.fr